Author's response to reviews

Title: Exploring health systems research and its influence on policy processes in low income countries

Authors:

Adnan A Hyder (ahyder@jhsph.edu)
Gerald Bloom (G.Bloom@ids.ac.uk)
Melissa Leach (M.Leach@ids.ac.uk)
Shamsuzzoha B Syed (ssyed@jhsph.edu)
David H Peters (dpeters@jhsph.edu)
Future Health Systems: Innovations for Equity (fhs@jhsph.edu)

Version: 8 Date: 26 October 2007

Author's response to reviews: see over
October 25, 2007

Dr Lolu da-Silva  
Assistant Editor  
BMC Public Health

Dear Dr. da-Silva:

I am pleased to submit a revised version of our manuscript based on the requests for changes to formatting in an email from the editorial production team dated October 25, 2007. I have responded to each of the items mentioned by the editorial production team. These editorial requests are listed below:

1. Please move the figure title and legend to the manuscript text. They should be placed in the text after the references.

2. Please remove the tables from the text and include them in the manuscript after the figure legends. Please place the title above the table.

3. Please do not use any table placement text, but ensure that all tables are mentioned within the main text, for example, see Table 1. The tables should be listed at the end of the manuscript in the order they have been mentioned in the text.

4. Tables should be made from cells not tabbed text/bullet points. Please reformat the tables.

There has been no need to make any changes apart from those suggested by the editorial production team. We look forward to the publication of the paper.

Thank you.

Sincerely,

Shamsuzzoha B. Syed, MD MPH DPH(Cantab)
Johns Hopkins University Bloomberg School of Public Health, General Preventive Medicine Residency Program